

Citation: He KY, Wang H, Cade BE, Nandakumar P, Giri A, Ware EB, et al. (2017) Rare variants in fox-1 homolog A (*RBFOX1*) are associated with lower blood pressure. PLoS Genet 13(3): e1006678. https://doi.org/10.1371/journal.pgen.1006678

Editor: Wei Chen, University of Pittsburgh, UNITED STATES

Received: October 10, 2016

Accepted: March 9, 2017

Published: March 27, 2017

Copyright: © 2017 He et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: CFS data are available at the International Sleep Genetic Epidemiology Consortium (ISGEC) (http:// sleepgenetics.org/downloads/). The summary statistics of replication data and all other data are within the paper and its Supporting Information files.

Funding: This work was supported by grants HG003054 from the National Human Genome Research Institute (to XZ), HL113338 (to SR, XL), HL098433 (to SR), HL46380 (to SR), HL007567 (to XZ), HL086694 (to AC), HL123677 (to NF) from RESEARCH ARTICLE

Rare variants in fox-1 homolog A (*RBFOX1*) are associated with lower blood pressure

Karen Y. He¹, Heming Wang¹, Brian E. Cade^{2,3}, Priyanka Nandakumar⁴, Ayush Giri⁵, Erin B. Ware^{6,7}, Jeffrey Haessler⁸, Jingjing Liang¹, Jennifer A. Smith⁷, Nora Franceschini⁹, Thu H. Le¹⁰, Charles Kooperberg⁸, Todd L. Edwards⁵, Sharon L. R. Kardia⁷, Xihong Lin¹¹, Aravinda Chakravarti⁴, Susan Redline^{2,3,12}, Xiaofeng Zhu¹*

1 Department of Epidemiology and Biostatistics, Case Western Reserve University, Cleveland, Ohio, United States of America, 2 Division of Sleep and Circadian Disorders, Brigham and Women's Hospital, Boston, Massachusetts, United States of America, 3 Division of Sleep Medicine, Harvard Medical School, Boston, Massachusetts, United States of America, 4 McKusick-Nathans Institute of Genetic Medicine, Johns Hopkins University School of Medicine, Baltimore, Maryland, United States of America, 5 Division of Epidemiology, Department of Medicine, Institute for Medicine and Public Health, Vanderbilt Genetics Institute, Vanderbilt University, Nashville, Tennessee, United States of America, 6 Biosocial Methods Collaborative, Institute for Social Research, University of Michigan, Ann Arbor, Michigan, United States of America, 7 Department of Epidemiology, School of Public Health, University of Michigan, Ann Arbor, Michigan, United States of America, 8 Division of Public Health Sciences, Fred Hutchinson Cancer Research Center, Seattle, Washington, United States of America, 9 Department of Epidemiology, UNC Gillings School of Global Public Health, Chapel Hill, North Carolina, United States of America, 10 Department of Medicine, Division of Nephrology, University of Virginia, Charlottesville, Virginia, United States of America, 11 Department of Biostatistics, Harvard T.H. Chan School of Public Health, Boston, Massachusetts, United States of America, 12 Division of Pulmonary, Critical Care, and Sleep Medicine, Beth Israel Deaconess Medical Center, Boston, Massachusetts, United States of America

* xiaofeng.zhu@case.edu

Abstract

Many large genome-wide association studies (GWAS) have identified common blood pressure (BP) variants. However, most of the identified BP variants do not overlap with the linkage evidence observed from family studies. We thus hypothesize that multiple rare variants contribute to the observed linkage evidence. We performed linkage analysis using 517 individuals in 130 European families from the Cleveland Family Study (CFS) who have been genotyped on the Illumina OmniExpress Exome array. The largest linkage peak was observed on chromosome 16p13 (MLOD = 2.81) for systolic blood pressure (SBP). Followup conditional linkage and association analyses in the linkage region identified multiple rare, coding variants in RBFOX1 associated with reduced SBP. In a 17-member CFS family, carriers of the missense variant rs149974858 are normotensive despite being obese (average $BMI = 60 \text{ kg/m}^2$). Gene-based association test of rare variants using SKAT-O showed significant association with SBP (p-value = 0.00403) and DBP (p-value = 0.0258) in the CFS participants and the association was replicated in large independent replication studies (N =57,234, p-value = 0.013 for SBP, 0.0023 for PP). RBFOX1 is expressed in brain tissues, the atrial appendage and left ventricle in the heart, and in skeletal muscle tissues, organs/tissues which are potentially related to blood pressure. Our study showed that associations of rare variants could be efficiently detected using family information.

the National Heart, Lung and Blood Institute, AG009740 from the National Institute on Aging (to SLRK), and HL121429 from the National Institutes of Health (to TLE). The Health and Retirement Study is supported by the National Institute on Aging (NIA U01AG009740), and funding for this analysis was also provided by the NIA (R03 AG046498). The genotyping was funded separately (RC2 AG036495, RC4 AG039029) and was conducted by the NIH Center for Inherited Disease Research (CIDR) at Johns Hopkins University. Genotyping quality control and final preparation of the data were performed by the University of Michigan School of Public Health. The Women's Health Initiative (WHI) program is funded by the National Heart, Lung, and Blood Institute, National Institutes of Health, US Department of Health and Human Services, through contracts HHSN268201100046C. HHSN268201100001C. HHSN268201100002C, HHSN268201100003C, HHSN268201100004C, and HHSN271201100004C. The BioVU dataset used in

HHSN271201100004C. The BioVU dataset used in the analyses described were obtained from Vanderbilt University Medical Center's BioVU, which is supported by institutional funding and by the Vanderbilt CTSA grant ULTR000445 from NCATS/NIH. Genotyping was funded by NIH grants RC2GM092618 from NIGMS/OD and U01HG004603 from NHGRI/NIGMS. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Hypertension is a risk factor for cardiovascular disease and the most important risk factor for stroke. Family studies suggest that hypertension related traits are heritable. Previous genome-wide association studies (GWAS) have identified multiple common blood pressure (BP) variants but these variants do not overlap with the linkage evidence observed from family studies. Rare variants have been suggested to play a substantial role and contribute to missing heritability of BP. In this study, linkage analysis identified 16p13 linked to SBP in a cohort of 517 white individuals in 130 families from the Cleveland Family Study (CFS). By combining linkage and association analyses, we searched for rare, coding variants that can explain the linkage evidence. Rare, coding variants within *RBFOX1* were associated with lower systolic (p-value = 0.00403) and diastolic (p-value = 0.0258) blood pressures, and explained significant amount of observed linkage evidence. We replicated the identified variants in four independent cohorts (with total sample size N = 57, 234) and further observed consistent evidence that rare *RBFOX1* variants are protectively associated with blood pressure traits. Our study clearly shows that family-based designs are powerful for identifying rare, coding variants underlying complex diseases.

Introduction

High blood pressure (BP) is a common condition associated with multiple health outcomes, including heart, brain, and kidney diseases [1, 2]. Previous studies have shown that BP is a genetically determined trait with estimated heritability of 30% to 60% [3, 4]. Multiple large genome-wide association studies (GWAS) meta-analysis and admixture mapping studies have identified over 190 genetic variants that explained only a small variation in BP [5-21].

For complex traits such as BP, rare variants are suggested to play a greater role in heritability than anticipated in the common disease-common variant hypothesis [22]. A Framingham Heart Study reported rare mutations in three renal salt handling genes causing large reductions in blood pressure and estimated that the overall prevalence of hypertension is reduced by about 1% because of the effects [23]. Linkage studies of family data can be used to uncover missing heritability and identify genetic markers linked to BP [24, 25]. However, the identified linkage regions from well-designed linkage studies such as the US Family Blood Pressure Program (FBPP) and the UK Medical Research Council British Genetics of Hypertension (BRIGHT) study [26–29] do not overlap with many BP loci identified by large BP GWAS of mostly unrelated individuals. In general, GWAS have good power to detect common variants of modest effect with attainable sample sizes, but less power for detecting rare variants with intermediate effect. In contrast, linkage analysis can have good power to detect multiple rare or lower frequency BP variants in a gene or region with relatively larger effect sizes [25]. Thus, we hypothesize that a linkage region observed in a family study, if not overlapping with the BP loci in reported GWAS, may harbor multiple rare or lower frequency BP variants.

Recently, many statistical approaches for rare variant association analyses have been developed for unrelated samples [30–34] and family data [35–38]. It has been suggested that rare or lower frequency variants can be enriched in families [35, 37], and therefore improving the statistical power for their identification. However, the existing rare variant association methods have not incorporated linkage evidence. In this study, we performed variance-component linkage analysis with BP traits, including systolic blood pressure (SBP), diastolic blood pressure (DBP), and pulse pressure (PP) in the Cleveland Family Study (CFS). We searched the published GWAS to examine whether there are reported BP variants in the linkage regions. Using the combined linkage and association analyses, we searched for potential functional variants that can explain linkage evidence and replicated the variants in independent cohorts.

Results

Table 1 presents the characteristics of white participants in the CFS data. S1 Fig presents the distributions of the residuals of SBP, DBP, and PP, which are all approximately normally distributed. Linkage analysis identified a peak (LOD = 2.81) on chromosome 16p13 linked to SBP (S2 Fig, Materials and Methods). Linkage analysis of further pruned markers using a R² threshold of 0.1 or modeling marker-marker linkage disequilibrium resulted in a slight decrease of LOD score in the same region (LOD = 2.30-2.42, S3 Fig). We selected a candidate region of 20cM with 2-LOD score drop for association analysis (Fig 1). This region did not overlap with published GWAS of BP variants. Therefore, we tested the hypothesis that the observed linkage evidence for SBP is due to the presence of multiple rare, coding variants in a gene(s) within the region.

The CFS was genotyped by an exome array, with most of the variants being coding variants. Within the linkage region, there are 454 exonic variants defined by the Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium that are genotyped on the exome array [39]. We identified 13 exonic variants (S1 Table) that satisfy the following filtering criteria: 1) either have a SBP association p-value ≤ 0.1 or absolute regression coefficient beta \geq 5; and 2) present at least twice in at least one family with a family-specific LOD score > 0.1. A risk score based on the 13 identified SNPs has an effect size of 0.948 \pm 0.135 for association with SBP in CFS. After adding this risk score as a covariate in linkage analysis, the MLOD score dropped from 2.809 to 1.055, suggesting that these 13 variants were able to account for most of the observed linkage evidence. To further assess the significance of this LOD drop, we sampled 1,000 independent SNPs from chromosomes other than chromosome 16. Hence, these SNPs should not contribute to the LOD score observed on chromosome 16. We performed linkage analysis with each of these 1,000 SNPs as a covariate in the linkage analysis. We calculated the differences between the original MLOD score and the MLOD scores of the 1,000 linkage analyses with a SNP as a covariate. The largest LOD score drop in these 1,000 linkage analyses was 0.347, suggesting the observed LOD score drop on the risk scores of 13 selected variants is statistically significant (p-value<0.001). Among these 13 exonic variants, two variants (rs149974858 and rs145873257) are present in *RBFOX1* and the remaining 11 variants are each in separate genes. When adjusting for the risk scores of rs149974858 and

Mean ± SD or N (%)
517
130
236 (45.6%)
46.9 ± 16.6
31.8 ± 8.55
123.7 ± 15.2
125.3 ± 16.6
73.4±9.70
74.5 ± 10.6
50.3 ± 12.2
50.8 ± 12.5

Table 1. Characteristics of white participants in the Cleveland Family Study.

https://doi.org/10.1371/journal.pgen.1006678.t001



Fig 1. Linkage region on chromosome 16 of white participants in CFS. Linkage peak on chromosome 16 for SBP. The linkage curves are plotted with (red curve) and without (blue curve) adjusting for the risk score defined by the 13 coding variants as a covariate. The positions of the 13 coding variants are listed under the linkage peak and above the genes.

https://doi.org/10.1371/journal.pgen.1006678.g001

PLOS GENETICS

rs145873257, we also observed a drop in LOD score (LOD = 1.97), which suggests that rs149974858 and rs145873257 account for a portion of the observed linkage evidence. The variant rs149974858 shows association evidence with SBP (p-value = 0.0016) in CFS. The minor allele frequency of rs149974858 is 0.0036 and only segregates within a 17-member family with family-specific LOD of 0.697 (Fig 2). This missense variant (c.112C>G) results in a proline to alanine substitution (p.Pro38Ala). Five members from this family carrying this rare missense mutation had on average lower SBP (carrier average = 117 mmHg, noncarrier average = 125 mmHg) but higher BMI than other family members (carrier average = 60 kg/ m^2 , noncarrier average = 31 kg/ m^2). A single SNP association test revealed that rs149974858 is also significantly associated with BMI in CFS (beta = -26.8±4.35, asymptotic p-value = 7.28E-10). The exonic variant rs145873257 segregates within a different family with family-specific LOD score of 0.215. A c.1072G>A base change resulted in a glycine to serine substitution (p. Gly358Ser). Six members of this family are heterozygous for the AT genotype. The estimated effect of this variant is protective in CFS, although it is not statistically significant (S1 Table).

Since both variants rs149974858 and rs148751394 consistently show a protective effect in two large families, we examined the other coding and rare variants in *RBFOX1* genotyped on the exome array. Two exonic variants, rs151214012 and rs145873257, and one rare, intronic variant rs2345080 are available in the exome array. These three variants show protective effect despite not satisfying the filtering criteria (S2 Table). Single SNP associations and annotations for all exome array variants of *RBFOX1* are provided in the S3 Table. Applying either the family-based burden or SKAT analysis, these five variants are significantly associated with SBP and DBP (Table 2).

We next sought the replication of the rare variants in *RBFOX1* in four large independent cohorts (ARIC, WHI, BioUV, and HRS) of whites for the traits SBP, DBP and PP. We specifically looked at the five rare variants of *RBFOX1* found in CFS and their associations with BP traits. Within the ARIC data, 4 of the 5 variants were present and all the 4 variants showed a consistent protective effect. For WHI, all 5 variants were present but only 1 variant (rs145873257) had a protective effect size. BioUV contained 4 out of 5 variants found in CFS; 2 variants were protective for SBP and 3 variants were protective for DBP. HRS contained all 5 variants found in CFS; 4 variants were protective for SBP and 3 variants were protective for DBP. Among the total 23 tests (CFS: 5, ARIC: 4, BioUV: 4, WHI: 5, HRS: 5) conducted across all cohorts, 16 of them were protective (p-value = 0.0173 based on binomial distribution), suggesting a consistent protective effect.

We next conducted a gene-based association analyses for *RBFOX1* and BP traits using exome array data from the CFS with weights Beta (1, 25). Burden, SKAT, and SKAT-O tests were performed using the 5 rare variants of *RBFOX1* found in CFS (Table 2). In CFS, the association between *RBFOX1* and SBP was found to be significant by the burden test (p-value = 0.00214) and SKAT-O (p-value = 0.00403), but not by SKAT (p-value = 0.0702). All three gene-based tests for DBP were statistically significant (burden test p-value = 0.0148, SKAT p-value = 0.0494, SKAT-O p-value = 0.0258). When we conducted gene-based analyses using only the 4 coding variants (rs149974858, rs148751394, rs151214012, rs145873257) identified in CFS, the associations are also significant for SBP and DBP (p-value<0.037).

The same gene-based analysis for rare variants was done for all four replication cohorts separately and the results were meta-analyzed (Table 2). Individually, the ARIC cohort had significant gene-based associations for SBP (burden test p-value = 0.00572, SKAT p-value = 0.00259, SKAT-O p-value = 0.00356) and PP (burden test p-value = 0.00140, SKAT p-value = 0.000273, SKAT-O p-value = 0.000392). After meta-analyzing the results for ARIC, WHI, BioVU, and HRS, we found significant associations for SBP (burden test p-value = 0.0172, SKAT pvalue = 0.00635, SKAT-O p-value = 0.0126) and PP (burden test p-value = 0.00377, SKAT pvalue = 0.00266, SKAT-O p-value = 0.00234).

We observed that the variant rs149974858 co-segregated with BMI in the 17-member CFS family. Subsequently, we performed linkage analysis for BMI in CFS on chromosome 16, after adjusting for gender, age, age², and PC1. We did not observe linkage evidence in this region





Fig 2. The CFS family carrying the protective rare variant rs149974858. A) The variant rs149974858 segregates with BP in a 17-member CFS family. Squares represent males and circles represent females. Half-filled subjects represent the carriers of the rare variant rs149974858. Grey subjects represent no information. Age of each subject is presented in parenthesis. B) The distribution of corresponding residuals of SBP, DBP, PP and BMI are presented.

https://doi.org/10.1371/journal.pgen.1006678.g002

	Methods	CFS	ARIC	WHI	BioVU	HRS	Meta of replication cohorts ^a
SBP							·
Rare variants	Burden	2.14E-3	5.72 E-3	6.68E-1	3.44E-1	6.95E-2	1.71E-2
	SKAT	7.02E-2	2.59E-3	6.71E-1	1.13E-1	1.19E-1	6.35E-3
	SKAT-O	4.03E-3	3.56E-3	8.14E-1	1.79E-1	1.15E-1	1.26E-2
DBP							
Rare variants	Burden	1.48E-2	3.28E-1	4.08E-1	9.44E-1	1.20E-1	3.98E-1
	SKAT	4.94E-2	4.61E-1	7.58E-1	3.98E-2	3.01E-1	2.05E-1
	SKAT-O	2.58E-2	4.75E-1	5.71E-1	6.42E-2	2.41E-1	2.05E-1
PP			·				
Rare variants	Burden	8.13E-2	1.40E-3	2.31E-1	2.52E-1	1.44E-1	3.77E-3
	SKAT	1.27E-1	2.73E-4	5.11E-1	4.23E-1	1.27E-1	2.66E-3
	SKAT-O	1.28E-1	3.92E-4	3.45E-1	3.80E-1	1.23E-1	2.34E-3
BMI							
Rare variants	Burden	4.24E-3	4.35E-1	N/A	6.40E-1	7.26E-1	7.84E-1
	SKAT	4.28E-4	6.50E-1	N/A	7.80E-1	7.26E-1	9.20E-1
	SKAT-O	7.32E-4	6.04E-1	N/A	8.07E-1	5.07E-1	8.34E-1
Number of rare variants		5	8	11	8	11	

Table 2. Results of gene-based analysis in discovery data and replication cohorts.

^a Meta-analysis of ARIC, WHI, BioVU, and HRS

https://doi.org/10.1371/journal.pgen.1006678.t002

(LOD = 0.721). The gene-based analysis of BMI using the same set of variants was only significant in CFS but not in any of the replication cohorts (Table 2).

Discussion

We performed a linkage analysis of BP traits using the families collected in CFS. The largest linkage peak identified is on 16p13 linked to SBP. The 16p13 region has been reported of linkage evidence with BP in two studies: the Victorian Family Heart Study [40], the extreme-sibpair study in Chinese adults [41]. In addition, the longitudinal change of BP in Mexican Americans [42], and the Hypertension Genetic Epidemiology Network Study in whites [43] reported linkage regions partially overlapped with the current study. The reported linkage evidence from multiple ethnic populations strongly suggests the linkage evidence on 16p13 is real. Linkage analysis using microsatellite markers was performed with 363 sib-pairs of CFS whites for a hypertensive status, adjusted for age, age², sex, BMI, and BMI². This analysis did not find linkage evidence in the 16p13 region. This is unsurprising because the power of using a binary hypertensive status is lower than that of quantitative phenotypes, such as SBP, DBP, or PP. In addition, hypertensive status was defined as either SBP \geq 140, DBP \geq 90, or taking antihypertensive medications and the sample size in the sib-pair analysis was smaller than the current study, all of which contribute to the lack of linkage evidence observed. Our study demonstrates that high-density SNP genotyping arrays are informative for detecting linkage signals.

We searched among published large GWAS studies of BP traits [5, 7–10] and did not identify any BP variants previously reported on 16p13, suggesting that multiple low frequency or rare variants with relatively large effect sizes are possibly contributing to the observed linkage evidence. We further assumed that variants with relatively large effect sizes are more likely to be coding and rare variants. Thus, we limited our search to only the coding and rare variants under the linkage peak genotyped on the exome array. By examining the associated variants that are able to account for the observed linkage evidence on 16p13, we were able to identify 13 exonic variants. Among these 13 variants, 2 of them fall in *RBFOX1*, which encodes for the ataxin-2 binding protein 1 (also known as *A2BP1*), and show a consistent protective effect for SBP in CFS. Gene-based analysis of the four available exonic variants and one rare intronic variant in *RBFOX1* are significantly associated with SBP and DBP (p-value = 0.00403, 0.0258, respectively) using SKAT-O. Replication analysis of the rare variants at the gene level (but not at the variant level) is also significant for SBP and PP in the meta-analysis of four large cohorts of whites with a total replication sample size N = 57,234.

This study also provides evidence that rare variants within *RBFOX1* are protective for BP traits among obese individuals. Among individuals of European ancestry within the CFS, 5 individuals within the same family carried the minor allele for rs149974858, the variant showing significantly protective effect with SBP by single SNP association test. All of the 5 individuals are morbidly obese (with average BMI of 60). However, their SBP (mean = 117) and DBP (mean = 78) are within the normotensive range. Ma et al. conducted a GWAS of BMI in Pima Indians using Affymetrix 100K array and identified two common variants in *RBFOX1* associated with BMI [44]. The same two variants could be replicated in non-overlapped Pima Indians but not in French Caucasians, Amish Caucasians, German Caucasians, or Native Americans [44]. In our analysis, we identified four exonic and one intronic rare variants in *RBFOX1* that are significantly associated with BMI but the association evidence could not be replicated (Table 2). Therefore, it is inconclusive whether *RBFOX1* is an obesity gene. In all our analysis, either linkage or association analysis, BMI is included as a covariate. Furthermore, no linkage evidence was found for BMI on chromosome 16, after adjusting for gender, age, age², and PC1. Our result indicates the *RBFOX1* contributes to BP variation independent of obesity, although we are unclear whether *RBFOX1* has a pleiotropic effect on both BP and obesity.

We also observed that the effect direction of single variant replication analysis in the four cohorts is not always consistent with that of CFS. However, 16 of the 23 tests were protective (p-value = 0.017), suggesting a consistent protective effect. Assuming a causal rare variant with an effect size equal to one quarter of the BP standard deviation, we estimate the probability of observing an opposite direction in a study to be 40.1%, which is consistent with 7 opposite directions among 18 single SNP replication tests in 4 replication cohorts.

It is interesting that all the four exonic variants in *RBFOX1* are either monomorphic or extremely rare in African ancestry populations (S4 Table). Furthermore, the BP admixture mapping analysis by Zhu et al. also suggest local ancestry in this region is associated with SBP and DBP; however, the evidence is not genome-wide significant [11]. Thus, our result suggests that the rare exonic variants in *RBFOX1* may contribute to a protective effect for hypertension and further work will be needed to establish whether the lack of these protective variants contribute to the disparity in hypertension occurrence and early age of onset between African Americans and whites.

To our knowledge, only one GWAS study so far has reported on the association of *RBFOX1* variants with blood pressure using human genotyping data. Wang et al. reported an association between rs1507023, a candidate SNP in *RBFOX1* involved in vitamin D metabolism and signaling, and SBP, DBP, and PP. Its association with blood pressure was significant before, but not after correction for multiple testing [45].

Under the linkage region of 16p13, there are 11 additional variants that either have an association test p-value less than 0.1 or an effect size larger than 5 mmHg in CFS. When we used the risk scores of these 11 variants as a covariate in linkage analysis, the MLOD dropped to 1.932, suggesting that there may be additional variants that contribute to linkage evidence in this region. However, the current exome array data is limited for further dissection of genes or variants contributing the linkage evidence. Whole genome sequencing data, including the







https://doi.org/10.1371/journal.pgen.1006678.g003

sequencing data from the Trans-Omics for Precision Medicine (TOPMed) Program (<u>https://</u>www.nhlbi.nih.gov/research/resources/nhlbi-precision-medicine-initiative/topmed), will be necessary to identify the genes contributing blood pressure variation in this region.

Gene expression data and previous studies have demonstrated that RNA splicing factor *RBFOX1* is important for heart and skeletal muscle development and function [46–48]. *RBFOX1* expression has been associated with cardiac hypertrophy and heart failure in mice models [49]. Gao et al. found that *RBFOX1* expression was significantly diminished in both mouse and human failing hearts [49]. We searched the GTEx database and *RBFOX1* is highly expressed in multiple human brain tissues, atrial appendage and left ventricle of the heart, as well as muscle skeletal tissues (Fig 3; http://www.gtexportal.org/home/gene/RBFOX1). Further biological studies are needed to establish the direct role of *RBFOX1* in regulating blood pressure.

Our study suggests that family-based linkage evidence can be extremely successful in searching for rare variants contributing to complex traits. In summary, we identified rare, exonic variants in *RBFOX1* that have a protective effect on BP traits, which can be important in searching new drugs for cardiovascular disease. However, it should be pointed out that association analysis was performed using variants available in the exome array of this study. The variants identified in *RBFOX1* may still be reflecting in LD with the causal variants to BP. While *RBFOX1* is expressed in multiple tissues that may relate to blood pressure, the mechanism underlying how this gene contributes to BP variation needs to be further studied. The identification of these rare coding variants will facilitate precision medicine in treating cardiovascular disease.

Materials and methods

The CFS is a family-based longitudinal study comprised of patients with laboratory-diagnosed sleep apnea, their family members, and neighborhood control families, as described before [50]. The data were analyzed anonymously at Case Western Reserve University. The CFS study was approved by Partners Human Research Committee with the proposal number 2011D001860. The analytical sample includes 517 white participants in 130 families who were genotyped with the Illumina OmniExpress Exome array, which includes both GWAS and exome variants (Table 1). Standard quality controls were performed, including checking Hardy-Weinberg equilibrium, Mendelian inconsistences, and verifying pedigree structure using the genetic markers by the software PLINK [51]. BP traits, including SBP and DBP were each determined following standardized guidelines using a calibrated sphygmomanometer [52]. Height and weight were directly measured and antihypertensive medications were ascertained by questionnaire. Data for this analysis were from the last available examination for each participant.

The samples used for replication analysis include five independent cohorts. We included 10,864 unrelated white participants from the Atherosclerosis Risk in Communities (ARIC) Study. The ARIC study is a prospective epidemiologic study designed to investigate the natural history and etiology of atherosclerosis (https://www2.cscc.unc.edu/aric/). There were 18,050 unrelated white participants from the Women's Health Initiative (WHI), a study of postmeno-pausal women focused on strategies for preventing heart disease, breast and colorectal cancer, and osteoporotic fractures [53, 54]. From the Vanderbilt University Biobank (BioVU), we included 18,977 unrelated white individuals. BioVU uses leftover blood samples collected during routine clinical care from consented individuals who visit the Vanderbilt University Medical Center [55]. Lastly, we included 9,343 unrelated white participants from the Health and Retirement Study (HRS). This is a longitudinal survey of a representative sample of Americans over the age of 50 [56–58].

SBP and DBP for an individual taking antihypertensive medication were imputed using the standard approach in literature, by adding 15 mmHg and 10 mmHg, respectively. Pulse pressure (PP) was calculated as the difference between imputed SBP and DBP [5].

Linkage analysis in CFS

We calculated residuals of SBP, DBP, and PP after adjusting for gender, age, age², body mass index (BMI; kg/m²) and the first principal component of genotype values in CFS separately. The residuals of these regressions were used for linkage analysis using the software MERLIN [59]. The principal components (PCs) were calculated using the software FamCC, which can be applied to family data [60]. Since the results were essentially the same for including the first PC or the first 10 PCs, we reported the linkage results including the first PC. We used the pairwise linkage disequilibrium (LD) pruning approach with a window size of 50 kb, step size of 5 variants, and R² threshold of 0.2. We also required a minor allele frequency (MAF) \geq 0.2. This resulted in 56,992 autosomal SNPs using PLINK [51]. Because marker-marker LD may result in biased linkage calculations, we performed linkage analysis by further reducing the R² threshold to 0.1 and by modeling the marker-marker LD using MERLIN [59]. Linkage analysis using MERLIN decomposes phenotypic variance into three parts: the variance contributes to the quantitative trait locus (σ_{QTL}^2), the variance contributes to the polygenetic effect (σ_G^2), and the variance contributes to the random effect (σ_E^2). It also tests the null hypothesis of no linkage $H_0: \sigma_{QTL}^2 = 0 \text{ vs. } H_A: \sigma_{QTL}^2 > 0.$

Identify coding variants accounting for linkage evidence

We examined exonic variants genotyped in the exome array in the region of 2-LOD score drop from the linkage peak. We performed the family-based association analysis for the exonic variants only in the 2-LOD drop region using the ASSOC package in S.A.G.E [61]. The familybased association analysis was conducted using a linear mixed model $y = \beta_0 + \beta_1 g + \delta + \epsilon$, where g is a genotype value vector, β_0 is the intercept, β_1 is the regression coefficient, $\delta \sim N(0, 2\Phi\sigma_G^2)$ where Φ is the kinship coefficient matrix and σ_G^2 is the polygenic variance, and $\epsilon \sim N(0, I\sigma_{\epsilon}^2)$ where σ_{ϵ}^2 is the random error. ASSOC applies the likelihood ratio test to test the null hypothesis of H_0 : $\beta_1 = 0$. For each of the variants, we first performed an association analysis with a BP trait using ASSOC and identified variants with either p-value ≤ 0.1 (marginal effect) or absolute regression coefficient beta ≥ 5 (large effect). We next estimated family-specific LOD scores and identified families with LOD score ≥ 0.1 . We kept the variants with association p-value ≤ 0.1 or absolute regression coefficient ≥ 5 , and that were present at least twice in at least one family with family-specific LOD score ≥ 0.1 .

We defined the risk score as $r_i = x_i^T \beta$, where β is the regression coefficients of the SNPs, and x_i is a vector of the number of risk alleles carried by individual *i* for these SNPs. Linkage analysis was further performed conditional on the risk scores.

SNP-set burden and SKAT test

We performed family-based burden and SKAT tests for CFS using the software famSKAT and for the replication cohorts using the R package SKAT [30, 32, 62]. The weight was set to $\sqrt{w_i} = Beta(MAF_i, 1, 25)$ as suggested to increase the weight of rare variants.

Supporting information

S1 Fig. The distributions of residuals of SBP, DBP, and PP in CFS. (TIFF)

S2 Fig. Linkage results of SBP, DBP, and PP in CFS. (TIF)

S3 Fig. Linkage analysis by additional LD-based pruning or modeling marker-marker LD for chromosome 16. Original (blue): pairwise LD pruning with a window size of 50 kb, step size of 5 variants, and R² threshold of 0.2; MAF \geq 0.2. Further pruning (purple): pairwise LD pruning with a window size of 50 kb, step size of 5 variants, and R² threshold of 0.1; MAF \geq 0.3. Cluster (red): modeling marker-marker LD using "—cluster" option with R² threshold of 0.1; all other parameters are the same as the original linkage analysis. (TIFF)

S1 Table. Single association analysis of 13 exonic variants within the linkage region. ^a Include both founders and nonfounders

(DOCX)

S2 Table. Single SNP association effect sizes and p-values for SBP, DBP, PP. (DOCX)

S3 Table. Single SNP association analysis for exome array variants of *RBFOX1* identified in CFS. ^a Include both founders and nonfounders (DOCX)

S4 Table. Minor allele frequencies ^a of *RBFOX1* rare coding variants in white, African-American, and African populations. ^a Include both founders and nonfounders.

^b Calculated by the weighted average of Cleveland Family Study, Atherosclerosis Risk in Communities, Women's Health Initiative, Vanderbilt University Biobank, and Health and Retirement Study.

^c Calculated by the weighted average of Atherosclerosis Risk in Communities, Family Blood Pressure Program, Africa America Diabetes Mellitus Study, and Howard University Family Study (H.W., unpublished data).

^d Calculated based on exome array of Nigeria data (H.W., unpublished data). (DOCX)

Author Contributions

Conceptualization: XZ.

Data curation: KYH HW PN AG EBW JH JL JAS BEC.

Formal analysis: KYH HW PN AG EBW JH JL JAS.

Funding acquisition: NF CK TLE AC SR XZ.

Investigation: BEC THL.

Methodology: XZ.

Project administration: XZ.

Resources: NF THL CK TLE SLRK XL AC SR XZ.

Software: KYH HW.

Supervision: XZ.

Visualization: KYH.

Writing – original draft: KYH XZ.

Writing - review & editing: KYH NF CK TLE AC SR XZ.

References

- 1. Sowers JR, Epstein M, Frohlich ED. Diabetes, Hypertension, and Cardiovascular Disease: An Update. Hypertension. 2001; 37(4):1053–9. PMID: <u>11304502</u>
- Hsu CY, McCulloch CE, Darbinian J, Go AS, Iribarren C. Elevated Blood Pressure and Risk of Endstage Renal Disease in Subjects Without Baseline Kidney Disease. Arch Intern Med. 2005; 165 (8):923–8. https://doi.org/10.1001/archinte.165.8.923 PMID: 15851645
- Kupper N, Willemsen G, Riese H, Posthuma D, Boomsma DI, de Geus EJC. Heritability of Daytime Ambulatory Blood Pressure in an Extended Twin Design. Hypertension. 2005; 45(1):80–5. https://doi. org/10.1161/01.HYP.0000149952.84391.54 PMID: 15557390
- van Rijn MJ, Schut AF, Aulchenko YS, Deinum J, Sayed-Tabatabaei FA, Yazdanpanah M, et al. Heritability of blood pressure traits and the genetic contribution to blood pressure variance explained by four blood-pressure-related genes. J Hypertens. 2007; 25(3):565–70. https://doi.org/10.1097/HJH. 0b013e32801449fb PMID: 17278972
- The International Consortium for Blood Pressure Genome-Wide Association Studies. Genetic variants in novel pathways influence blood pressure and cardiovascular disease risk. Nature. 2011; 478 (7367):103–9. https://doi.org/10.1038/nature10405 PMID: 21909115
- 6. Franceschini N, Reiner AP, Heiss G. Recent Findings in the Genetics of Blood Pressure and Hypertension Traits. American Journal of Hypertension. 2011; 4(24):392–400.

- Ganesh SK, Tragante V, Guo W, Guo Y, Lanktree MB, Smith EN, et al. Loci influencing blood pressure identified using a cardiovascular gene-centric array. Human molecular genetics. 2013; 22(8):1663–78. https://doi.org/10.1093/hmg/dds555 PMID: 23303523
- Levy D, Ehret GB, Rice K, Verwoert GC, Launer LJ, Dehghan A, et al. Genome-wide association study of blood pressure and hypertension. Nature Genetics. 2009; 41(6):677–87. https://doi.org/10.1038/ng. 384 PMID: 19430479
- Newton-Cheh C, Johnson T, Gateva V, Tobin MD, Bochud M, Coin L, et al. Genome-wide association study identifies eight loci associated with blood pressure. Nature Genetics. 2009; 41(6):666–76. https://doi.org/10.1038/ng.361 PMID: 19430483
- Tragante V, Barnes MR, Ganesh SK, Lanktree MB, Guo W, Franceschini N, et al. Gene-centric metaanalysis in 87,736 individuals of European ancestry identifies multiple blood-pressure-related loci. American journal of human genetics. 2014; 94(3):349–60. https://doi.org/10.1016/j.ajhg.2013.12.016 PMID: 24560520
- Zhu X, Young JH, Fox E, Keating BJ, Franceschini N, Kang S, et al. Combined admixture mapping and association analysis identifies a novel blood pressure genetic locus on 5p13: contributions from the CARe consortium. Human molecular genetics. 2011; 20(11):2285–95. https://doi.org/10.1093/hmg/ ddr113 PMID: 21422096
- Wang YJ, Tayo BO, Bandyopadhyay A, Wang H, Feng T, Franceschini N, et al. The association of the vanin-1 N131S variant with blood pressure is mediated by endoplasmic reticulum-associated degradation and loss of function. PLoS Genet. 2014; 10(9):e1004641. https://doi.org/10.1371/journal.pgen. 1004641 PMID: 25233454
- Zhu X, Cooper RS. Admixture mapping provides evidence of association of the VNN1 gene with hypertension. PLoS One. 2007; 2(11):e1244. https://doi.org/10.1371/journal.pone.0001244 PMID: 18043751
- Cho YS, Go MJ, Kim YJ, Heo JY, Oh JH, Ban HJ, et al. A large-scale genome-wide association study of Asian populations uncovers genetic factors influencing eight quantitative traits. Nat Genet. 2009; 41 (5):527–34. https://doi.org/10.1038/ng.357 PMID: 19396169
- Kato N, Loh M, Takeuchi F, Verweij N, Wang X, Zhang W, et al. Trans-ancestry genome-wide association study identifies 12 genetic loci influencing blood pressure and implicates a role for DNA methylation. Nat Genet. 2015; 47(11):1282–93. https://doi.org/10.1038/ng.3405 PMID: 26390057
- Kato N, Takeuchi F, Tabara Y, Kelly TN, Go MJ, Sim X, et al. Meta-analysis of genome-wide association studies identifies common variants associated with blood pressure variation in east Asians. Nat Genet. 2011; 43(6):531–8. https://doi.org/10.1038/ng.834 PMID: 21572416
- Zhu X, Luke A, Cooper RS, Quertermous T, Hanis C, Mosley T, et al. Admixture mapping for hypertension loci with genome-scan markers. Nat Genet. 2005; 37(2):177–81. https://doi.org/10.1038/ng1510 PMID: 15665825
- Zhu X, Feng T, Tayo BO, Liang J, Young JH, Franceschini N, et al. Meta-analysis of correlated traits via summary statistics from GWASs with an application in hypertension. Am J Hum Genet. 2015; 96(1):21– 36. https://doi.org/10.1016/j.ajhg.2014.11.011 PMID: 25500260
- Surendran P, Drenos F, Young R, Warren H, Cook JP, Manning AK, et al. Trans-ancestry meta-analyses identify rare and common variants associated with blood pressure and hypertension. Nat Genet. 2016; 48(10):1151–61. https://doi.org/10.1038/ng.3654 PMID: 27618447
- Liu C, Kraja AT, Smith JA, Brody JA, Franceschini N, Bis JC, et al. Meta-analysis identifies common and rare variants influencing blood pressure and overlapping with metabolic trait loci. Nat Genet. 2016; 48(10):1162–70. https://doi.org/10.1038/ng.3660 PMID: 27618448
- Ehret GB, Ferreira T, Chasman DI, Jackson AU, Schmidt EM, Johnson T, et al. The genetics of blood pressure regulation and its target organs from association studies in 342,415 individuals. Nat Genet. 2016; 48(10):1171–84. https://doi.org/10.1038/ng.3667 PMID: 27618452
- Doris PA. The Genetics of Blood Pressure and Hypertension: The Role of Rare Variation. Cardiovascular Therapeutics. 2011; 29(1):37–45. <u>https://doi.org/10.1111/j.1755-5922.2010.00246.x</u> PMID: 21129164
- Ji W, Foo JN, O'Roak BJ, Zhao H, Larson MG, Simon DB, et al. Rare independent mutations in renal salt handling genes contribute to blood pressure variation. Nat Genet. 2008; 40(5):592–9. https://doi. org/10.1038/ng.118 PMID: 18391953
- Eichler EE, Flint J, Gibson G, Kong A, Leal SM, Moore JH, et al. Missing heritability and strategies for finding the underlying causes of complex disease. Nature Reviews Genetics. 2010; 11(6):446–50. https://doi.org/10.1038/nrg2809 PMID: 20479774
- Manolio TA, Collins FS, Cox NJ, Goldstein DB, Hindorff LA, Hunter DJ, et al. Finding the missing heritability of complex diseases. Nature. 2009; 461(7265):747–53. https://doi.org/10.1038/nature08494 PMID: 19812666

- Province MA, Kardia SL, Ranade K, Rao DC, Thiel BA, Cooper RS, et al. A meta-analysis of genomewide linkage scans for hypertension: the National Heart, Lung and Blood Institute Family Blood Pressure Program. Am J Hypertens. 2003; 16(2):144–7. PMID: 12559682
- Caulfield M, Munroe P, Pembroke J, Samani N, Dominiczak A, Brown M, et al. Genome-wide mapping of human loci for essential hypertension. Lancet. 2003; 361(9375):2118–23. Epub 2003/06/27. https:// doi.org/10.1016/S0140-6736(03)13722-1 PMID: 12826435
- Wu X, Kan D, Province M, Quertermous T, Rao DC, Chang C, et al. An updated meta-analysis of genome scans for hypertension and blood pressure in the NHLBI Family Blood Pressure Program (FBPP). Am J Hypertens. 2006; 19(1):122–7. Epub 2006/02/08. https://doi.org/10.1016/j.amjhyper. 2005.07.010 PMID: 16461203
- Cooper RS, Luke A, Zhu X, Kan D, Adeyemo A, Rotimi C, et al. Genome scan among Nigerians linking blood pressure to chromosomes 2, 3, and 19. Hypertension. 2002; 40(5):629–33. PMID: 12411454
- Lee S, Wu MC, Lin X. Optimal tests for rare variant effects in sequencing association studies. Biostatistics. 2012; 13(4):762–75. Epub 2012/06/16. <u>https://doi.org/10.1093/biostatistics/kxs014</u> PMID: 22699862
- Madsen BE, Browning SR. A groupwise association test for rare mutations using a weighted sum statistic. PLoS Genet. 2009; 5(2):e1000384. https://doi.org/10.1371/journal.pgen.1000384 PMID: 19214210
- Wu MC, Lee S, Cai T, Li Y, Boehnke M, Lin X. Rare-variant association testing for sequencing data with the sequence kernel association test. Am J Hum Genet. 2011; 89(1):82–93. https://doi.org/10.1016/j. ajhg.2011.05.029 PMID: 21737059
- Neale BM, Rivas MA, Voight BF, Altshuler D, Devlin B, Orho-Melander M, et al. Testing for an unusual distribution of rare variants. PLoS Genet. 2011; 7(3):e1001322. https://doi.org/10.1371/journal.pgen. 1001322 PMID: 21408211
- Li B, Leal SM. Methods for detecting associations with rare variants for common diseases: application to analysis of sequence data. Am J Hum Genet. 2008; 83(3):311–21. <u>https://doi.org/10.1016/j.ajhg.</u> 2008.06.024 PMID: 18691683
- Zhu X, Feng T, Li Y, Lu Q, Elston RC. Detecting rare variants for complex traits using family and unrelated data. Genetic epidemiology. 2010; 34(2):171–87. https://doi.org/10.1002/gepi.20449 PMID: 19847924
- Feng T, Elston RC, Zhu X. Detecting rare and common variants for complex traits: sibpair and odds ratio weighted sum statistics (SPWSS, ORWSS). Genetic epidemiology. 2011; 35(5):398–409. https://doi.org/10.1002/gepi.20588 PMID: 21594893
- Epstein MP, Duncan R, Ware EB, Jhun MA, Bielak LF, Zhao W, et al. A statistical approach for rare-variant association testing in affected sibships. Am J Hum Genet. 2015; 96(4):543–54. <u>https://doi.org/10. 1016/j.ajhg.2015.01.020</u> PMID: 25799106
- Chen R, Wei Q, Zhan X, Zhong X, Sutcliffe JS, Cox NJ, et al. A haplotype-based framework for groupwise transmission/disequilibrium tests for rare variant association analysis. Bioinformatics. 2015; 31 (9):1452–9. https://doi.org/10.1093/bioinformatics/btu860 PMID: 25568282
- Grove ML, Yu B, Cochran BJ, Haritunians T, Bis JC, Taylor KD, et al. Best practices and joint calling of the HumanExome BeadChip: the CHARGE Consortium. PLoS One. 2013; 8(7):e68095. <u>https://doi.org/ 10.1371/journal.pone.0068095</u> PMID: 23874508
- Harrap SB, Wong ZY, Stebbing M, Lamantia A, Bahlo M. Blood pressure QTLs identified by genomewide linkage analysis and dependence on associated phenotypes. Physiol Genomics. 2002; 8(2):99– 105. https://doi.org/10.1152/physiolgenomics.00069.2001 PMID: 11875187
- Xu X, Rogus JJ, Terwedow HA, Yang J, Wang Z, Chen C, et al. An extreme-sib-pair genome scan for genes regulating blood pressure. Am J Hum Genet. 1999; 64(6):1694–701. <u>https://doi.org/10.1086/</u> 302405 PMID: 10330357
- 42. Rutherford S, Cai G, Lopez-Alvarenga JC, Kent JW, Voruganti VS, Proffitt JM, et al. A chromosome 11q quantitative-trait locus influences change of blood-pressure measurements over time in Mexican Americans of the San Antonio Family Heart Study. Am J Hum Genet. 2007; 81(4):744–55. <u>https://doi.org/10.1086/521151</u> PMID: <u>17846999</u>
- Shi G, Gu CC, Kraja AT, Arnett DK, Myers RH, Pankow JS, et al. Genetic effect on blood pressure is modulated by age: the Hypertension Genetic Epidemiology Network Study. Hypertension. 2009; 53 (1):35–41. https://doi.org/10.1161/HYPERTENSIONAHA.108.120071 PMID: 19029486
- 44. Ma L, Hanson RL, Traurig MT, Muller YL, Kaur BP, Perez JM, et al. Evaluation of A2BP1 as an Obesity Gene. Diabetes. 2010; 59(11):2837–45. https://doi.org/10.2337/db09-1604 PMID: 20724578
- 45. Wang L, Chu A, Buring JE, Ridker PM, Chasman DI, Sesso HD. Common Genetic Variations in the Vitamin D Pathway in Relation to Blood Pressure. American Journal of Hypertension. 2014; 27 (11):1387–95. https://doi.org/10.1093/ajh/hpu049 PMID: 24688000

- 46. Frese KS, Meder B, Keller A, Just S, Haas J, Vogel B, et al. RNA splicing regulated by RBFOX1 is essential for cardiac function in zebrafish. Journal of Cell Science. 2015; 128(16):3030–40. https://doi. org/10.1242/jcs.166850 PMID: 26116573
- Pedrotti S, Giudice J, Dagnino-Acosta A, Knoblauch M, Singh RK, Hanna A, et al. The RNA-binding protein Rbfox1 regulates splicing required for skeletal muscle structure and function. Human Molecular Genetics. 2015; 24(8):2360–74. https://doi.org/10.1093/hmg/ddv003 PMID: 25575511
- Baron AD, Brechtel-Hook G, Johnson A, Hardin D. Skeletal muscle blood flow. A possible link between insulin resistance and blood pressure. Hypertension. 1993; 21(2):129–35. Epub 1993/02/01. PMID: 8428775
- 49. Gao C, Ren S, Lee J-H, Qiu J, Chapski DJ, Rau CD, et al. RBFox1-mediated RNA splicing regulates cardiac hypertrophy and heart failure. Journal of Clinical Investigation. 2015; 126(1):195–206. <u>https://doi.org/10.1172/JCI84015 PMID: 26619120</u>
- Tishler PV, Larkin EK, Schluchter MD, Redline S. Incidence of sleep-disordered breathing in an urban adult population: the relative importance of risk factors in the development of sleep-disordered breathing. JAMA. 2003; 289(17):2230–7. https://doi.org/10.1001/jama.289.17.2230 PMID: 12734134
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MA, Bender D, et al. PLINK: a tool set for wholegenome association and population-based linkage analyses. Am J Hum Genet. 2007; 81(3):559–75. Epub 2007/08/19. https://doi.org/10.1086/519795 PMID: 17701901
- Kirkendall WM BA, Epstein FH, Freis ED. Recommendations for human blood pressure determination by sphygmomanometers. Subcommittee of the AHA Postgraduate Education Committee. Circulation. 1980; 62(5):1146A–55A. PMID: 7418174
- Design of the Women's Health Initiative Clinical Trial and Observational Study. Controlled Clinical Trials. 1998; 19(1):61–109. PMID: 9492970
- Anderson GL, Manson J, Wallace R, Lund B, Hall D, Davis S, et al. Implementation of the women's health initiative study design. Annals of Epidemiology. 2003; 13(9):S5–S17.
- Roden DM, Pulley JM, Basford MA, Bernard GR, Clayton EW, Balser JR, et al. Development of a largescale de-identified DNA biobank to enable personalized medicine. Clin Pharmacol Ther. 2008; 84 (3):362–9. https://doi.org/10.1038/clpt.2008.89 PMID: 18500243
- Juster FS, R. An Overview of the Health and Retirement Study. Journal of Human Resources. 1995; 30:S7–S56.
- Weinstein MV, JW; Wachter KW. Committee on Advances in Collecting and Utilizing Bilogical Indicators and Genetic Information in Social Science Surveys. Biosocial Surveys: National Academies; 2007. p. 78.
- Sonnega A, Faul JD, Ofstedal MB, Langa KM, Phillips JW, Weir DR. Cohort Profile: the Health and Retirement Study (HRS). Int J Epidemiol. 2014; 43(2):576–85. <u>https://doi.org/10.1093/ije/dyu067</u> PMID: 24671021
- Abecasis GR, Cherny SS, Cookson WO, Cardon LR. Merlin—rapid analysis of dense genetic maps using sparse gene flow trees. Nat Genet. 2002; 30(1):97–101. Epub 2001/12/04. https://doi.org/10. 1038/ng786 PMID: 11731797
- Zhu X, Li S, Cooper RS, Elston RC. A unified association analysis approach for family and unrelated samples correcting for stratification. Am J Hum Genet. 2008; 82(2):352–65. Epub 2008/02/07. https:// doi.org/10.1016/j.ajhg.2007.10.009 PMID: 18252216
- Elston RC, Gray-McGuire C. A review of the 'Statistical Analysis for Genetic Epidemiology' (S.A.G.E.) software package. Human genomics. 2004; 1(6):456–9. https://doi.org/10.1186/1479-7364-1-6-456 PMID: 15607000
- Chen H, Meigs JB, Dupuis J. Sequence kernel association test for quantitative traits in family samples. Genetic epidemiology. 2013; 37(2):196–204. https://doi.org/10.1002/gepi.21703 PMID: 23280576